Author's response to reviews

Title: Differences in the quality of interpersonal care in complementary and conventional medicine

Authors:

André Busato (andre.busato@memcenter.unibe.ch)
Beat Künzi (beat.kuenzi@swisspep.ch)

Version: 6 Date: 2 November 2010

Author's response to reviews: see over
Resubmission of an article to BMC Complementary and Alternative Medicine

Dear Sirs

Thank you for accepting our paper entitled „Differences in the quality of interpersonal care in complementary and conventional medicine“ submitted for publication in BMC Complementary and Alternative Medicine.

We revised the paper according to the comments of the Editorial Production Team. The following manuscript changes were made:

- We rewrote the abstract in narrative format
- We copied the new version of the abstract into the submission system
- We added a sentence indicating that all authors read and approved the final manuscript to the respective section of the manuscript.
- We renumbered the tables and changed the references in the text accordingly.
- We completed references 11, 20 and 28
- We removed the additional file from the manuscript

We now resubmit the article for the final steps of publication.

With best regards

André Busato